Author's response to reviews

Title: Recurrence of hepatitis C virus during leucocytopenia and spontaneous clearance after recovery from cytopenia: a case report

Authors:

Norbert H Gruener (Norbert.Gruener@med.uni-muenchen.de)
Bijan Raziorrouh (Bijan.Raziorrouh@med.uni-muenchen.de)
Maria-Christina Jung (Maria-Christina.Jung@med.uni-muenchen.de)

Version: 7 Date: 29 November 2007

Author's response to reviews:

Formatting changes have be done as requested:
1. Author list exactly as in the submission system
2. Affiliation as requested
3. e-mail addresses of all authors are given
4. Consent section is now placed after Authors contribution
5. Acknowledgment at the end of the manuscript text, directly preceding the reference list
6. Table in portrait layout